

G OPEN ACCESS

Citation: Trippler L, Ame SM, Hattendorf J, Juma S, Abubakar S, Ali SM, et al. (2021) Impact of seven years of mass drug administration and recrudescence of *Schistosoma haematobium* infections after one year of treatment gap in Zanzibar: Repeated cross-sectional studies. PLoS Negl Trop Dis 15(2): e0009127. https://doi.org/ 10.1371/journal.pntd.0009127

Editor: Antonio Montresor, World Health Organization, SWITZERLAND

Received: September 18, 2020

Accepted: January 12, 2021

Published: February 12, 2021

Copyright: © 2021 Trippler et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data that support the findings of this study from 2011/12-2017 are openly available in ClinEpiDB. The dataset "Study: SCORE Zanzibar *S. haematobium* Cluster Randomized Trial" can be found at https:// clinepidb.org/ce/app/record/dataset/DS_ eddb4757ba. The data that support the findings of this study from 2018-2020 are within the manuscript and its supporting information. RESEARCH ARTICLE

Impact of seven years of mass drug administration and recrudescence of *Schistosoma haematobium* infections after one year of treatment gap in Zanzibar: Repeated cross-sectional studies

Lydia Trippler^{1,2}, Shaali Makame Ame³, Jan Hattendorf^{1,2}, Saleh Juma⁴, Salum Abubakar⁴, Said Mohammed Ali³, Fatma Kabole⁴, David Rollinson⁵, Stefanie Knopp^{1,2}*

Swiss Tropical and Public Health Institute, Basel, Switzerland, 2 University of Basel, Basel, Switzerland,
Public Health Laboratory—Ivo de Carneri, Pemba, United Republic of Tanzania, 4 Neglected Diseases
Program, Ministry of Health, Zanzibar, United Republic of Tanzania, 5 Department of Life Sciences, Natural
History Museum, London, United Kingdom

* s.knopp@swisstph.ch

Abstract

Background

Considerable progress towards the elimination of urogenital schistosomiasis was made by the Zanzibar Elimination of Schistosomiasis Transmission project from 2012 till 2016, when biannual praziquantel mass drug administration (MDA) alone or with additional snail control or behaviour change interventions were implemented. Annual MDA was continued in 2017 and 2018, but not in 2019, imposing a 16-month treatment gap. We monitored the *Schistosoma haematobium* prevalence from 2012 till 2020 and assessed recrudescence patterns with focus on 2020.

Methodology

Repeated cross-sectional surveys were conducted from 2011/12 till 2020 in 90 communities and 90 schools in Zanzibar. Annually, around 4,500 adults and up to 20,000 schoolchildren were surveyed. The *S. haematobium* prevalence was detected by urine filtration and reagent strips. In 2020, risk factors for infection were investigated using generalized estimated equation models.

Principal findings

In adults, the apparent *S. haematobium* prevalence was 3.9% in 2011 and 0.4% in 2020. In schoolchildren, the prevalence decreased from 6.6% in 2012 to 1.2% in 2019 with vicissitudes over the years. Prominent recrudescence of infection from 2.8% in 2019 to 9.1% (+225%) in 2020 was observed in 29 schools with historically moderate prevalences (\geq 10%). Compared with 2019, reinfection in 2020 was particularly striking in boys aged

Funding: This study was funded by the University of Georgia Research Foundation Inc., which is funded by the Bill & Melinda Gates Foundation (https://www.gatesfoundation.org/) for the Schistosomiasis Consortium for Operational Research and Evaluation (SCORE; https://score. uga.edu/) projects (prime award no. 50816, subaward no. RR374-053/4893206 to DR). SK received financial support by sub-award no. RR374-053/4893196 and via a direct grants from the Bill & Melinda Gates Foundation (Investment IDs: OPP1191423 and OPP1198086). Praziguantel was donated by the World Health Organization, Geneva, Switzerland and treatment implementation was supported by the Schistosomiasis Control Initiative, Imperial College, London, United Kingdom. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

9–16 years. Being male was a risk factor for infection in 2020 (adults: odds ratio (OR): 6.24, 95% confidence interval (95% CI): 1.96–19.60; schoolchildren: OR: 2.06, 95% CI: 1.52–2.78). Living near to a natural freshwater body significantly increased the odds of infection in adults (OR: 2.90, CI: 1.12–7.54).

Conclusions/Significance

After 11 rounds of MDA over 7 years and a 16-month treatment gap, the urogenital schistosomiasis prevalence considerably rebounded in hotspot areas. Future elimination efforts in Zanzibar should focus on re-intensifying MDA plus additional interventions in hotspot areas. In low-prevalence areas, the strategy might be adapted from MDA to targeted surveillanceresponse.

Author summary

Schistosomiasis is a neglected tropical disease caused by parasitic blood flukes of the genus Schistosoma. On the Zanzibar islands, United Republic of Tanzania, interventions to eliminate urogenital schistosomiasis commenced in 2012. From 2012 to 2016, the population was treated biannually with praziquantel and, additionally, some areas received mollusciciding against the intermediate host snail, or educational measures for behavior change. Mass drug administration (MDA) with praziquantel was continued annually in 2017 and 2018, but not in 2019. As a result of the interventions, the overall S. haematobium prevalence was reduced to 0.4% in adults and 3.4% in schoolchildren in 2020. However, in some areas, the MDA gap in 2019 resulted in a considerable rebound of infections. The recrudescence in 2020 was particularly striking for boys aged 9-16 years. In general, in 2020, male participants had higher odds of infection than females. Adults living near to a natural freshwater body also showed an increased risk of S. haematobium infection. Future elimination efforts in Zanzibar should focus on reintensifying elimination interventions, including MDA, snail control and behavior change in hotspot areas. In low-prevalence areas, the strategy might be adapted from MDA to targeted interventions, such as surveillance-response.

Introduction

The neglected tropical disease (NTD) schistosomiasis is endemic in 78 countries worldwide and, according to estimates from 2017, it is responsible for about 1.4 million disabilityadjusted life years annually [1,2]. In their new road map for NTDs 2021–2030, the World Health Organization (WHO) highlights the global elimination of schistosomiasis as a public health problem as target for 2030 [3]. Elimination as public health problem is therein defined as <1% proportion of heavy intensity infections [3]. According to mathematical modelling as well as multi-country and multi-year studies, elimination of schistosomiasis as public health problem can be achieved relatively quickly [4–6]. Once countries or areas have achieved this goal, they likely wish to move further down the elimination road, aiming for the interruption of schistosomiasis transmission, defined as zero incidence of infection.

Clearly, as long as interruption of transmission has not been achieved, recrudescence of transmission and disease remains a threat. Recrudescence of transmission can easily occur due

to the asexual reproduction of the parasite in its intermediate host snail and the release of thousands of cercariae into natural freshwater, which are then able to infect the next definitive human host [7,8]. Hence, as long as some, even very few people remain infected with schistosomes, the parasite's lifecycle can be perpetuated and the risk of recrudescence remains [9]. Also after the transmission of the parasite was successfully interrupted in an area or country, transmission can be reintroduced as long as the intermediate host snails thrive and *Schistosoma* eggs are released into the snails' freshwater habitat. To avoid recrudescence in areas where interruption of transmission has not yet been achieved, control and elimination efforts need to be maintained. To prevent reintroduction of schistosomiasis in areas considered post-elimination settings, a sensitive detection and timely treatment of cases through effective surveillance-response mechanisms is necessary [3]. Regarding the clonal reproduction of the parasite in its intermediate host snail, the detection of the parasite in the snail by xenomonitoring will constitute an important part of surveillance activities [7,8].

Interventions to control the morbidity caused by urogenital schistosomiasis in Zanzibar started in the 1980's when S. haematobium was highly prevalent on the islands Pemba and Unguja [10–13]. Control interventions were enforced by large-scale mass drug administration (MDA) campaigns in schools implemented in the early 2000s [14–16]. These interventions paved the way for one of the first projects to eliminate urogenital schistosomiasis from an area in sub-Sahara Africa: the Zanzibar Elimination of Schistosomiasis Transmission (ZEST) project, implemented in Pemba and Unguja from 2011 till 2017 [17,18]. Over the course of ZEST, considerable progress towards the elimination of urogenital schistosomiasis was made [6,19]. During the ZEST project, MDA with praziquantel was implemented biannually in schools and whole communities across the islands Pemba and Unguja. In addition to biannual MDA, snail control and behavioural change interventions were implemented in 30 selected shehias (smallest administrative areas) in Pemba and Unguja, respectively, as part of a cluster randomized trial supported by the Schistosomiasis Consortium for Operational Research and Evaluation (SCORE) [17]. The impact of the interventions was monitored in annual cross-sectional surveys in 90 study schools and shehias from 2012 until 2017. The interventions reduced the S. haematobium prevalence in Zanzibar from 3.9% to 1.5% in 20-55 year old adults, and from 6.6% to 1.9% in 9–12 year old schoolchildren from 2012 to 2017. In 2017, heavy S. haematobium infection intensities (>50 eggs per 10 ml urine) occurred in 0.1% of 20-55 year old adults and in 0.5% of 9- to 12-year old schoolchildren [19]. Hence, within the ZEST project, urogenital schistosomiasis was eliminated as public health problem from most schools and communities in Zanzibar in 2017 and the S. haematobium prevalence was significantly reduced [6,19]. While the SCORE trial was concluded in early 2017, MDA in schools and shehia communities continued under the lead of the Zanzibar NTD Program, and was implemented annually in 2017 and 2018. No MDA was conducted in 2019 due to procurement issues.

We aimed to monitor the *S. haematobium* prevalence in children and adults in Zanzibar from 2012 till 2020 and to assess recrudescence patterns and risk factors for infection across all age groups with a particular focus on the year 2020, after the 16-month treatment gap.

Methods

Ethics statement

The SCORE cluster randomized intervention trial, including annual cross-sectional surveys to assess the *S. haematobium* prevalence in 90 study schools and shehias in Pemba and Unguja from 2012 till 2018 was approved by the Zanzibar Medical Research Ethics Committee in Zanzibar, United Republic of Tanzania (ZAMREC, reference no. ZAMREC 0003/Sept/011), the

Ethikkommission beider Basel (EKBB) in Basel, Switzerland (reference no. 236/22) and the Institutional Review Board of the University of Georgia in Athens, Georgia, United States of America (project no. 2012-10138-0). The trial is registered in the International Standard Randomized Controlled Trial Number register (ISRCTN; ISRCTN48837681). The cross-sectional surveys conducted in 2019 and 2020 obtained ethical approval from the Zanzibar Health Research Institute in Zanzibar (ZAHRI), United Republic of Tanzania (reference no. NO. ZAHREC/02/June/2019/36) and the Ethics Committee Northwest and Central Switzerland (EKNZ) in Basel, Switzerland (Project ID: Req-2019-00524). The surveys are registered in ISRCTN (ISRCTN17656730). All individuals participating in the cross-sectional surveys from 2011/12 until 2020 were informed in detail about the objectives and procedures of the studies and provided written informed consent for their participation. In the case of children below the age of 18, written consent was obtained from their parents or guardians.

Study site

The Zanzibar islands (Pemba and Unguja) are a semi-autonomous part of the United Republic of Tanzania. Both islands are located in the Indian Ocean, about 30 km off the coast from the east African mainland [20]. The islands are divided into 11 districts, which are again subdivided into 388 small administrative areas, called shehias [21]. The estimated population of Zanzibar for 2019 was 1.6 million with a population density of 669 person/km² in Unguja and 503 person/km² in Pemba. The proportion of the population aged 0–17 years was 49.0% [21]. Public primary schools in Zanzibar contain the grades 1–6 and some schools additionally include a nursery school. The net enrolment rate in primary schools was 83.2% in 2014/15 and the percentage of people being employed or full-time student was 67.8% [22]. Among the total population of Zanzibar, 28.9% had access to an improved water supply source in 2014/15 and 59.0% had access to an own improved toilet facility in 2015/16 [22,23].

Interventions for schistosomiasis elimination in Zanzibar

From 2012 to 2016, MDA was conducted biannually in Pemba and Unguja by the NTD Program of the Zanzibar Ministry of Health within the ZEST project [17]. The inhabitants of all shehias, excluding those located in the South district and in parts of the Urban district in Unguja, were offered praziquantel via community-wide treatment (CWT) or school-based treatment (SBT) approaches. During CWT, trained community drug distributors offered praziquantel (40 mg/kg) in a door-to-door approach to all community members aged >3 years that did not receive praziquantel in the same treatment round via SBT, or were not pregnant or severely sick, using a dose pole [24,25]. CWT was conducted biannually (twice a year) from April 2012 to November 2016 on both islands [6]. In SBT, trained teachers provided directlyobserved praziquantel treatment to the children attending school on the day of treatment using a dose pole [24,25]. SBT was implemented biannually, from November 2013 until November 2016, with exception of 2014 where SBT was conducted only once a year in Pemba, but not at all in Unguja.

In addition to biannual MDA, 15 randomized shehias in Pemba and Unguja, respectively, received snail control interventions and 15 randomized shehias on each island received behaviour change interventions as part of the SCORE cluster randomized trial from 2012 till 2016. These interventions are described elsewhere in detail [6,17,19].

Once the SCORE trial and respective interventions were concluded in early 2017, the Zanzibar NTD Program continued the implementation of MDA. In 2017 and 2018, both CWT and SBT were conducted once a year. In 2019, no MDA was implemented, due to difficulties in procurement.

Study design and participants

The SCORE study implemented from 2011/12 until 2017 was designed as a 5-year cluster randomized trial. The impact of biannual MDA alone or in combination with snail control or behavior change interventions was monitored annually with cross-sectional surveys in a total of 90 shehias and 90 schools in Pemba and Unguja, respectively [17]. Details of the design of the SCORE study, including the justification of the number of participants, are described elsewhere [17,19]. In brief, the SCORE study population included annually ~50 adults aged 20- to 55 years in each of the 90 shehias that were enrolled in parasitological community-based surveys (CBS) that also included a questionnaire component. Moreover, ~100 schoolchildren aged 9- to 12-years from grade 3 and 4 in each of the 90 study schools were enrolled annually in parasitological school-based surveys (SBS) that also included a questioning component. Additionally, in the years 2012 and 2017, ~100 schoolchildren from grade 1 in each of the 90 study schools were enrolled in the SBS [19].

After the SCORE study was concluded in early 2017, the annual cross-sectional surveys continued with the same design and sample sizes for children aged 9- to 12-years in the SBS and for adults aged 20- to 55-years in the CBS in 2018, 2019, and 2020 to monitor the impact of annual MDA in 2017 and 2018 and the treatment gap in 2019 on *S. haematobium* infections.

In 2018, only a SBS including schoolchildren aged 9- to 12-years from grade 3 and 4 as well as children from grade 1 was conducted, but no CBS.

In 2019 and 2020, we were interested not only in the age groups targeted by SCORE, but aimed to investigate the extent of *S. haematobium* infections in a broader age group. Hence, in the CBSs in the years 2019 and 2020, not only 50 individuals aged 20–55 years as in SCORE, but 70 individuals aged 15–100 years were sampled per shehia. Moreover, in the SBS, the original sample size of ~100 children from grades 3 and 4 in each study school was kept as in SCORE, and additionally 20 children each from nursery, grade 1, 2, 5, and 6 were added.

The baseline CBS was conducted in November and December 2011. The baseline SBS was conducted from February till April 2012. All other CBS and SBS were conducted between January and May every year. Unfortunately, in 2020, the CBS and SBS were not completed due to the Covid-19 pandemic. Therefore, in 2020, the surveys were limited to 68 shehias and 67 corresponding schools (Fig 1).

Field procedures

Each year before the surveys commenced, the community leader (sheha) and head teacher of each study shehia and school, respectively, were invited to a meeting, where the procedures and aims of the forthcoming survey were explained and results from the previous survey were presented. The meeting participants were requested to inform their shehia population and school, respectively, about the purpose of the survey.

In the CBS, the survey teams visited each shehia for one day. In the years 2011–2017, data of ~50 adults per shehia were collected. In the years 2019 and 2020, data of ~70 adolescents and adults (aged 15–100 years) per shehia were collected. The randomization and selection of households and participants is described elsewhere in detail [17]. In all years, the study procedure was explained in Kiswahili to the members of the selected households by a trained field-worker and the selected participants were asked to provide written consent by signing the informed consent form with their signature or thumbprint. Subsequently, they were invited to produce and submit a fresh urine sample and to answer questions from a pre-tested questionnaire. Besides the collection of demographic data, participants were asked about behaviors that might put them at risk of *S. haematobium* infection, for example their participation in the last MDA round, contact with natural freshwater, access to clean water, and travel history.



Fig 1. Flowchart of study design.

https://doi.org/10.1371/journal.pntd.0009127.g001

In the SBSs, each school was visited by the study team for two subsequent working days. The randomization of schoolchildren for participation is explained elsewhere in detail [17]. In brief, in each grade eligible for participation, girls and boys lined up stratified by sex, and every third child was selected until the requested number was reached. In 2012–2020, ~120 children from grades 3 and 4 were selected. In addition, in 2012, 2017 and 2018, ~120 children from grade 1 were selected. Finally, in 2019 and 2020, ~25 children each from grades 1, 2, 5 and 6 as well as from nursery were selected. The additional grades were included in 2019 and 2020 to obtain a better picture on the age profile of *S. haematobium* infections. The study procedure was explained in lay terms in Kiswahili to the enrolled children. Upon registration, the children were asked about their demographic data, their travel history, and in years with preceding MDA also about their participation in the last SBT round. All enrolled children received an information- and a consent-sheet for their parents or legal guardians to read and sign. On the next day, all children that submitted an informed consent form signed by their parents or legal guardians were given a plastic container (100 ml) to provide their fresh own urine sample.

All urine samples collected in CBS and SBS were produced between 10:00 and 14:00. The samples from Unguja were transferred to the laboratory of the NTD Program and all urine samples collected in Pemba were taken to the Public Health Laboratory–Ivo de Carneri (PHL-IdC).

Laboratory procedures

The urine samples were examined in the laboratories on the day of collection. First, the samples were investigated for visible macrohematuria using a color chart and for microhematuria

using reagent strips (Siemens Health Care AG, Zürich, Switzerland). Subsequently, each urine sample with a volume of at least 10 ml was vigorously shaken and filtered through a polycarbonate filter (Sefar Ltd., Bury, United Kingdom), which was then examined under a microscope for the presence of *S. haematobium* eggs by a well-experienced laboratory technician. The grade of macrohematuria and microhematuria, respectively, as well as the number of *S. haematobium* eggs for each participant was recorded on a laboratory form.

Outcomes

The primary outcome variable was the *S. haematobium* infection status in study participants from Pemba and Unguja in each cross-sectional survey.

The secondary analyses covered the annual *S. haematobium* prevalence and infection intensity as well as microhematuria levels among 20-to 55-year old adults and 9- to 12-year old children in the years 2011/12 to 2020 at aggregated and shehia/school level. Moreover, we assessed the age-prevalence distribution of *S. haematobium* infections in the years 2019 and 2020, stratified by sex, and including study participants of all ages. Finally, we determined the odds of infection in relation to demographic and behavioral factors in 2020.

Data management and statistical analyses

In all cross-sectional surveys from 2011/12 till 2020, data collected during registration in the SBS and laboratory examinations of the CBS and SBS were recorded on paper forms by the field enumerators and laboratory technicians, respectively, and subsequently double entered in electronic spreadsheets (Excel 2010, Microsoft) by data entry clerks in Zanzibar. In the years 2011/12 till 2019, paper questionnaire data collected in the CBS were entered into EpiInfo version 3.5.4 (Centers for Disease Control and Prevention, Atlanta, United States of America) by local staff in Zanzibar. In 2020, CBS questionnaire data were collected with the software Open Data Kit (https://opendatakit.org/) installed on Samsung Galaxy 4 tablets (Samsung Electronics, Seoul, South Korea). All data were cleaned and analyzed with StataIC 16 (StataCorp., Texas, United States of America) and R version 4.0.3 (www.rpoject.org). The data that support the findings of this study from 2011/12-2017 are openly available in ClinEpiDB. The dataset "Study: SCORE Zanzibar *S. haematobium* Cluster Randomized Trial" can be found at https:// clinepidb.org/ce/app/record/dataset/DS_eddb4757ba. The data that support the findings of this study from 2018–2020 are within the manuscript and its supporting information (S1 Data and S1 Dictionary).

All study participants with a written informed consent and results of the urine filtration and/or reagent strip examination were included in the statistical analyses. A participant was considered *S. haematobium* positive if at least one egg per 10 ml urine was detected by the urine filtration method. Infections were stratified into light intensity (1–49 eggs per 10 ml urine) and heavy intensity (\geq 50 eggs per 10 ml urine) infections, according to WHO guidelines [26]. In the absence of a urine filtration result, a participant was considered *S. haematobium*-positive if the urine was microhematuria-positive, as detected by reagent strips. The intensity of microhematuria was graded into negative, trace, 1, 2 or 3, in line with the color code provided by the reagent strip's manufacturer.

For the determination of the annual *S. haematobium* prevalence and infection intensity at aggregated and cluster level, only data from the 20-to 55-year old adults and from the 9-to 12-year old schoolchildren collected in the years 2011/12 to 2020 were included in the analyses. These restrictions were applied across the years, since during the SCORE study from 2011/12-2017, only data from these age groups were collected and we aimed to keep the annual data comparable.

For the differentiation of hotspot *versus* low prevalence areas, we defined a hotspot area as a shehia or school, which had a point prevalence of \geq 10% in at least one of the study years and a low prevalence area as a shehia or school with point prevalences <10% throughout all study years from 2011/12 till 2020. Maps containing information about the spatial distribution of prevalence were created with R version 4.0.3. Coordinates of schools were collected with a handheld Garmin GPSMAP 62sc device (Garmin, Kansas City, USA). Shape files of shehias were provided by the Zanzibar Health Management Information System to the NTD Program of the Zanzibar Ministry of Health.

For the determination of the age profile of *S. haematobium* infections, participants of all ages of the 2019 and 2020 surveys were included in the analyses. Generalized estimating equation (GEE) models with exchangeable correlation structure were applied to estimate the odds ratios (OR) for *S. haematobium* infection. Separate models were run for the CBS and SBS, respectively, in 2020. The following "risk factors" were investigated as explanatory variables: sex (binary variable), age (categorical variable), travel history (binary variable), place of birth (categorical variable), living close to a natural freshwater body (binary variable; in CBS only), using water from natural freshwater bodies (binary variable; in CBS only), having a tap in or near the house (binary variable; in CBS only), having a well in or near the house (binary variable; in CBS only), and school grade (categorical variable; in SBS only). Besides the variables "place of birth" (excluded in both surveys due to collinearity), "use of water from natural freshwater bodies" (excluded in CBS due to collinearity), and "age" (excluded in SBS due to collinearity), all listed variables were included in the multivariable GEE models. In the models, 95% confidence intervals (CI) were used, and GEE with robust standard errors to account for clustering. Stratified by CBS and SBS, either the shehias (CBS) or the schools (SBS) were included in the model as clusters.

Results

Study flow and participant characteristics

The baseline CBS in 2011 was conducted in 89 among 90 allocated shehias in Pemba and Unguja; one allocated shehia was mistakenly not surveyed. In the CBSs implemented from 2013 to 2017 and in 2019, all 90 allocated shehias were surveyed. In 2020, the CBS was interrupted due to the Covid-19 pandemic, and only 68 among 90 allocated shehias were surveyed (Fig 1). In each of the CBS implemented from 2011 to 2017, around 4,400 adults aged 20–55 years participated. In the CBSs in 2019 and 2020, 6,300 and 4,749 individuals aged 15 years and older were enrolled.

The baseline SBS in 2012 was carried out in 88 among 90 allocated schools in Pemba and Unguja; two allocated schools were mistakenly not surveyed. In the SBSs implemented from 2013 to 2015, all 90 allocated schools were surveyed. Due to a school closing in 2016, the SBSs in 2016 till 2019 were conducted in 89 schools. Due to the Covid-19 pandemic and the associated interruption of survey activities in Zanzibar in March 2020, the SBS in 2020 was limited to 67 schools (Fig 1). In the SBSs conducted from 2012 to 2018, around 9,700 schoolchildren aged 9–12 years participated each year. In 2019 and 2020, a total of 19,559 and 14,544 children, respectively, were enrolled. Characteristics of the study participants from each CBS and SBS, respectively, are presented in Table 1.

Change in annual *S. haematobium* prevalence and intensity and microhematuria levels

Fig 2 shows that in the CBSs implemented from 2011 till 2020, the overall *S. haematobium* prevalence decreased constantly among the 20–55 year old adults from 3.9% in 2011 to 0.4% in 2020. An exception was 2014, when the prevalence increased by 0.5%-points (from 3.0% to

Survey	Characteristic	Stratification	2011/12	2013	2014	2015	2016	2017	2018	2019	2020
CBS	Shehias - n		89	90	90	90	90	90	0	90	68
	Participants with outcome data - n		3974	4378	4462	4469	4474	4487	0	6300	4749
	Sex - n (%)	Female	2835 (71)	2671 (61)	2821 (63)	2478 (55)	2395 (54)	2133 (48)	0	3327 (53)	3010 (63)
		Male	1136 (29)	1707 (39)	1641 (37)	1991 (45)	2079 (46)	2354 (52)	0	2973 (47)	1739 (37)
	Age group - n	\leq 19 year old adults	0	0	0	0	0	0	0	1205	627
		20-55 year old adults	3974	4378	4462	4469	4474	4487	0	4742	3717
		\geq 56 year old adults	0	0	0	0	0	0	0	344	372
SBS	Schools - n		88	90	90	90	89	89	89	89	67
	Participants with outcome data - n		8931	9379	9595	9803	9725	10632	9662	19559	14544
	Sex - n (%)	Female	4664 (52)	5041 (54)	5098 (53)	5093 (52)	5122 (53)	5308 (50)	4903 (51)	9814 (50)	7407 (51)
		Male	4267 (48)	4338 (46)	4497 (47)	4710 (48)	4603 (47)	5324 (50)	4759 (49)	9745 (50)	7137 (49)
	Age group - n	\leq 8 year old children	0	0	0	0	0	0	0	4622	3606
		9-12 year old children	8931	9379	9595	9803	9725	10632	9662	12017	9081
		\geq 13 year old children	0	0	0	0	0	0	0	2906	1856

Table 1.	Characteristics of the study participants fr	om community-based survey	(CBS) and school-based survey	(SBS) by year.
----------	--	---------------------------	-------------------------------	----------------

3.5%) compared with the previous year. The relative difference in the prevalence from 2011 to 2020 was -91%. Heavy intensity infections were found in 0.3% of participants at baseline in 2011, reached a maximum of 0.5% in 2014, and declined to zero in 2020.

Microhematuria levels in 20–55 year old adults were higher than the *S. haematobium* prevalence in all study years. From 2011 till 2020, the overall microhematuria among the 20–55 year old adults decreased from 12.2% in 2011 to 6.3% in 2020. A higher overall microhematuria level than in 2011 was observed in 2014 (15.3%). The highest grading of microhematuria (3) ranged between 1.0% and 3.7% and no decreasing trend in heavy microhematuria was observed over the study period.

In the SBS, a decreasing trend of the overall *S. haematobium* prevalence among the 9–12 year old children was observed from 2012 (6.6%) to 2019 (1.2%), with some inconsistent upturns in certain years (Fig 2). The relative difference in the prevalence from 2012 to 2019 was -82%. At baseline in 2012, heavy intensity infections were detected in 1.8% of participants and decreased to 0.3% in 2019 (-83%). A prominent rebound in the overall *S. haematobium* infections and intensity was observed after the 16-month treatment gap in 2020, when the prevalence of infections rose back from 1.2% to 3.4% (relative difference +183%) and heavy intensity infections rebounded from 0.3% to 0.8% (relative difference +167%).

Microhematuria levels in 9–12 year old schoolchildren were higher than the *S. haematobium* prevalence in all study years, except in 2013. In line with the *S. haematobium* prevalence, a decreasing trend of the overall microhematuria levels was observed from 2012 (9.5%) till 2019 (4.0%). The relative difference in the percentage of the overall microhematuria from 2012 to 2019 was -58%. At baseline in 2012, heavy microhematuria levels (3) were detected in 2.6% of participants and decreased to 1.2% in 2019 (-54%). A rebound in the overall microhematuria level was observed after the 16-month treatment gap in 2020, when the overall levels rose back from 4.0% to 5.2% (relative difference +30%) and heavy microhematuria rebounded from 1.2% to 2.5% (relative difference +108%).

Spatial heterogeneity of S. haematobium prevalence

The geographical distribution of the shehias and schools, and the *S. haematobium* prevalence in 20–55 year-old adults and 9–12 year-old children from 2011/12 to 2020 is indicated in Fig 3.



Fig 2. Prevalence and intensity of *S. haematobium* infections, microhematuria levels, and implementation of MDAs in Zanzibar from 2011/12 until 2020. The data show the *S. haematobium* prevalence and infection intensity, the levels of microhematuria, and implementation of MDA per annual quartile (Q). A: 20- to 55-year old adults; B: 9- to 12-year old schoolchildren. Blue: light intensity infection prevalence; red: heavy intensity infection prevalence. Very light brown: microhematuria graded trace; light brown: microhematuria graded 2; dark brown: microhematuria graded 3. Dark green: community-wide treatment (CWT) only; dark and light green: community-wide (CWT) and school-based treatment (SBT).

A total of 71 shehias and 61 schools showed a prevalence <10% throughout all study years and were therefore defined as low prevalence areas, whereas 19 shehias and 29 schools crossed the 10% line in one or several years and were therefore considered hotspot areas. Fig 4 shows that the hotspot areas were more prone to recrudescence of *S. haematobium* infections and heavy intensity infections in one year or the other, and mainly responsible for the strong rebound observed after the 16-month treatment gap in 2020. While the rebound of the overall *S. haematobium* prevalence in schools from 2019 to 2020 was +183% (from 1.2% to 3.4%), the



Fig 3. Maps indicating prevalence of *S. haematobium* infections from 2011/12 till 2020 in spatial clusters in Pemba (A) and Unguja (B). Data are *S. haematobium* prevalence in 45 schools/shehias (clusters) per island, stratified by population group and island. A: Pemba; B: Unguja. Data are presented per year starting from 2011/12 (12 o'clock position) to 2020. Inner circle: 20- to 55-year old adults; outer circle: 9- to 12-year old schoolchildren. The colors show the prevalence from dark green as lowest prevalence to dark red as highest prevalence, and grey indicates missing. The * represents a hotspot school and/or community. Maps containing information about the spatial distribution of prevalence were created with R version 4.0.3. Coordinates of schools were collected with a handheld Garmin GPSMAP 62sc device (Garmin, Kansas City, USA). Shape files of shehias were provided by the Zanzibar Health Management Information System to the Neglected Diseases Program of the Zanzibar Ministry of Health.



Fig 4. Prevalence of *S. haematobium* **infections stratified by infection intensity levels in Zanzibar from 2011/12 until 2020 per spatial cluster in Pemba (A) and Unguja (B).** Data are proportion of *S. haematobium* infection intensity in schoolchildren and adults in each surveyed cluster on each island. A: Shehias and schools in Pemba; B: Shehias and schools in Unguja. X-axis: year of survey (2011/12 to 2020); y-axis: proportion infection intensity. Dark blue: heavy intensity infection among 20- to 55-year old adults; light blue: light intensity infection among 20- to 55-year old adults. Dark red: heavy intensity infection among 9- to 12-year old schoolchildren; light red: light intensity infection among 9- to 12-year old schoolchildren.

rebound in hotspot areas was +225% (from 2.8% to 9.1%) and the rebound in the low prevalence areas was +75% (from 0.4% to 0.7%).

Age prevalence distribution

Fig 5 shows the age prevalence distribution of *S. haematobium* infections stratified by sex and in relation to the sampling effort in 2019 (Fig 5A) and 2020 (Fig 5B). In 2019, the *S. haematobium* infection levels were below 2.0% in both, male and female students, and community members. The prevalence was significantly higher in boys aged 8 to 16 years, compared with female students of the same age. No participant older than 42 years was tested *S. haematobium*-positive. In 2020, after the 16-month treatment gap, a prominent increase in prevalence was observed in 9–16 year-old schoolchildren of both sexes. Among the whole study population and all ages younger than 60 years, male participants had higher levels of infections than female participants did.

Risk factors for S. haematobium infection in 2020

In the CBS conducted in 2020, men had significantly higher chances of harboring a *S. haematobium* infection compared with women (OR: 6.24, 95% CI: 1.96–19.60, p = 0.002) (Fig 6A). Adults aged 20–55 years had significantly lower odds of having a *S. haematobium* infection than community members aged below 20 years (OR: 0.30, 95% CI: 0.13–0.69, p = 0.004). A *S. haematobium* infection was significantly associated with living in close proximity to a natural freshwater body (OR: 2.90, 95% CI: 1.12–7.54, p = 0.029). Having a tap in or near the house, travel outside the home island over the past 6 months, and praziquantel treatment within the last two years resulted in non-significant lower odds of *S. haematobium* infection.

In the SBS, boys had significantly higher chances of having a *S. haematobium* infection compared with girls (OR: 2.06, 95% CI: 1.52–2.78, p < 0.001) (Fig 6B). In each of the school grades 1 to 6, children had a higher chance of harboring a *S. haematobium* infection compared with children who attended nursery school, with children from grade 1 showing the lowest odds ratio (OR: 8.49, 95% CI: 1.89–38.16, p = 0.005) and children from grade 4 showing the highest odds ratio (OR: 15.70, 95% CI: 4.30–57.35, p < 0.001). A recent travel history outside the home island resulted in non-significantly lower odds of *S. haematobium* infection.

Discussion

Considerable progress towards the elimination of urogenital schistosomiasis was made by the ZEST project from 2012 till 2017, when biannual praziquantel MDA alone or with additional snail control or behaviour change interventions were applied. Annual MDA was continued across Zanzibar in 2017 and 2018, but not in 2019. We monitored the *S. haematobium* prevalence and microhematuria levels from 2012 until 2020 and assessed recrudescence patterns and risk factors for infection across all age groups with a particular focus on the year 2020, after the 16-month treatment gap.

We found that the apparent overall *S. haematobium* prevalence among adults decreased from 3.9% at baseline in 2011 to 0.35% in 2020. Overall microhematuria levels in adults decreased from 12.2% in 2011 to 6.3% in 2020. Microhematuria levels in adults were remarkably higher than the *S. haematobium* prevalence measured by egg output by urine filtration microscopy and also than the microhematuria levels in children. This observation might indicate that either many adults suffer from chronic morbidity due to urogenital schistosomiasis that is not reflected by the number of eggs excreted in urine, or, microhematuria in adults might be caused by reasons unrelated to schistosomiasis such as urinary tract infections, bladder stones, sickle cell anemia, or pregnancy or menstrual bleeding in females [27–29].



Fig 5. *S. haematobium* prevalence across all ages, stratified by sex, after 11 rounds of MDA in 2019 (A) and a 16-month treatment gap in 2020 (B). The blue (male) and red (female) dots in the upper part (pos) of each figure, each represent a *S. haematobium*-positive urine sample from an individual of a certain age (x-axis). The blue (male) and red (female) dots in the lower part (neg) of each figure, each represent a *S. haematobium*-negative urine sample prom an individual of a certain age (x-axis). The blue (male) and red (female) and red (female) lines in the middle part of each figure represent the predicted *S. haematobium* age-prevalence. The shading around the lines represent 95% confidence bands.

The apparent overall *S. haematobium* prevalence among schoolchildren decreased from 6.6% at baseline in 2012 to 3.4% in 2020 and microhematuria levels from 9.5% to 5.2%, respectively. However, an even lower *S. haematobium* prevalence of 1.2% and microhematuria levels

.

A			
Covariates	N	Prev [%]	
Unguja	2512	0.3	÷
Pemba	2237	0.8	
Female	3010	0.2	÷
Male	1739	1.2	
Aged <20 years	627 2717	1.6	•
Aged 20-55 years	3/1/	0.3	
Aged >55 years	634	0.5	
Tap in/near the house	4115	0.5	
No well in/near the house	1678	0.2	•
Well in/near the house	3071	0.7	
No travel in last 6 months	3879	0.6	÷
Travel in last 6 months	855	0.2	
No treatment in last 2 years	881	1.1	÷
Treatment in last 2 years	2133	0.5	
Not living near freshwater body	3041	0.3	*
Living near freshwater body	1706	1	• • • • • • • • • • • • • • • • • • •
			0.1 0.5 2.0 5.0 20.0
в			
Coveriete e	N	Drov [9/1	
Covariates	N 7453	Prev [%]	
Covariates Unguja	N 7453	Prev [%] 3.4	
Covariates Unguja Pemba	N 7453 7087	Prev [%] 3.4 2.4	• • • • • • • • • • • • • • • • • • •
Covariates Unguja Pemba Female	N 7453 7087 7405	Prev [%] 3.4 2.4 1.9	
Covariates Unguja Pemba Female Male	N 7453 7087 7405 7135	Prev [%] 3.4 2.4 1.9 3.9	
Covariates Unguja Pemba Female Male Nursery school	N 7453 7087 7405 7135 1249	Prev [%] 3.4 2.4 1.9 3.9 0.2	· · · · · · · · · · · · · · · · · · ·
Covariates Unguja Pemba Female Male Nursery school Grade 1	N 7453 7087 7405 7135 1249 1415	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9	
Covariates Unguja Pemba Female Male Nursery school Grade 1 Grade 2	N 7453 7087 7405 7135 1249 1415 1506	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9 3	
Covariates Unguja Pemba Female Male Nursery school Grade 1 Grade 2 Grade 3	N 7453 7087 7405 7135 1249 1415 1506 3661	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9 3 3.3	
Covariates Unguja Pemba Female Male Nursery school Grade 1 Grade 2 Grade 3 Grade 4	N 7453 7087 7405 7135 1249 1415 1506 3661 3683	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9 3 3.3 3.3	
Covariates Unguja Pemba Female Male Nursery school Grade 1 Grade 2 Grade 3 Grade 4 Grade 5	N 7453 7087 7405 7135 1249 1415 1506 3661 3683 1523	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9 3 3.3 3.3 3.6 3.2	
Covariates Unguja Pemba Female Male Nursery school Grade 1 Grade 2 Grade 2 Grade 3 Grade 4 Grade 5	N 7453 7087 7405 7135 1249 1415 1506 3661 3683 1523	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9 3 3.3 3.6 3.6 3.2	
Covariates Unguja Pemba Female Male Nursery school Grade 1 Grade 2 Grade 2 Grade 3 Grade 4 Grade 5 Grade 6	N 7453 7087 7405 7135 1249 1415 1506 3661 3683 1523 1503	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9 3 3.3 3.3 3.6 3.2 2.8	
Covariates Unguja Pemba Female Male Nursery school Grade 1 Grade 2 Grade 2 Grade 3 Grade 4 Grade 5 Grade 6 No travel in last 6 months	N 7453 7087 7405 7135 1249 1415 1506 3661 3683 1523 1503 13522	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9 3 3.3 3.3 3.6 3.2 2.8 3	
Covariates Unguja Pemba Female Male Nursery school Grade 1 Grade 2 Grade 2 Grade 3 Grade 4 Grade 5 Grade 6 No travel in last 6 months Travel in last 6 months	N 7453 7087 7405 7135 1249 1415 1506 3661 3683 1523 1503 1503 13522 1018	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9 3 3.3 3.6 3.2 2.8 3 1.8	
Covariates Unguja Pemba Female Male Nursery school Grade 1 Grade 2 Grade 2 Grade 3 Grade 4 Grade 5 Grade 6 No travel in last 6 months Travel in last 6 months	N 7453 7087 7405 7135 1249 1415 1506 3661 3683 1523 1503 13522 1018	Prev [%] 3.4 2.4 1.9 3.9 0.2 1.9 3 3.3 3.6 3.2 2.8 3 1.8	

Adjusted Odds Ratio

Fig 6. Multivariable analysis of risk factors for *S. haematobium* infections in adults (A) and schoolchildren (B) in Zanzibar in 2020. The figure shows the odds ratios for a *S. haematobium* infection adjusted for different risk factors. Grey dots indicate the reference categories. N: total number of participants analyzed per group; Prev [%]: *S. haematobium* prevalence per group.

https://doi.org/10.1371/journal.pntd.0009127.g006

of 4.0% were observed in schoolchildren in 2019. In 2020, the considerable rebound in the overall prevalence and also infection intensity was caused by certain hotspot areas, while most areas had very low prevalences throughout the study period and also after the 16-month treatment gap. The hotspot areas showed an unstable and undulating *S. haematobium* prevalence

pattern over the years from 2011/12 to 2020. Rapid reinfection after praziquantel treatment and the persistence of hotspots are well-known challenges for the sustainable control and elimination of schistosomiasis [4,30-33].

Most of the hotspot areas in Pemba and Unguja were not geographically isolated but neighbored with other hotspot areas. Clustering of *S. haematobium* infections in specific areas has also been observed in other studies, for example in Kenya [30,34]. The likelihood of reinfection is strongly dependent on the environment of regions and the force of transmission [9,35,36]. For example, an earlier study in some of the hotspot areas in Unguja conducted in 2014 showed that there were considerably more natural freshwater bodies in total and also more natural fresh water bodies containing the intermediate host snail transmitting *S. haematobium* compared with investigated low-prevalence areas [37].

In our study, the rebound in *S. haematobium* (re)infections from 2019 to 2020 was most prominent in children aged 8–16 year-old, and in boys. The latter observation is in line with another study from Senegal, which reports a higher *S. haematobium* reinfection rate in boys compared with girls after treatment with praziquantel [38]. However, the full age-profile of *Schistosoma* infections remains poorly researched, particularly in populations that received several rounds of MDA. We showed that in 2019, after 11 rounds of MDA in Zanzibar, infection levels across all age groups from four to 80 years were extremely low and a "real" age-prevalence distribution was inexistent. The "typical" age-prevalence curve, reported in reviews with a peak of *S. haematobium* infections occurring in children aged 8–15 years and a subsequent decrease and stabilization of a low prevalence in adult age [39,40], reformed quickly in the 16-month treatment gap in Zanzibar until early 2020.

Our risk factor analyses in 2020 indicated that men and boys had significantly higher odds of a *S. haematobium* infection than women and girls (OR: 6.24 and OR: 2.06, respectively). The observations that male individuals are more likely to harbor schistosomiasis is in line with previous surveys in Zanzibar and elsewhere [18,41–44]. Male individuals may be more engaged in behaviors that expose them to *S. haematobium* contaminated freshwater, such as swimming, playing and fishing [41,45,46]. In Pemba and Unguja, where Islam is the predominant religion, men also often use the rivers and ponds for ablution before their prayers. School aged children might get in frequent contact with contaminated freshwater when doing household chores such as washing dishes or clothes at the rivers and ponds near their home, or while engaging in leisure activities such as swimming and playing [47,48]. While boys might expose themselves more frequently to water than girls and thus acquire infections more often and quickly, we also observed that in line with their religious beliefs, girls frequently keep their clothes on while bathing or swimming, which might impose a certain barrier for cercarial penetration.

The risk factor analysis of adults in the CBS showed significantly elevated odds of *S. haematobium* infection for individuals who reported to live in close proximity to natural freshwater bodies (OR: 2.90). This observation confirms earlier reports from Zanzibar, where the proximity of households or schools to freshwater bodies containing the intermediate host snails was associated with *S. haematobium* infections [37,45]. Travel outside the home island, for example to the sister island, or mainland Tanzania, or other African countries, was not a significant risk factor in our study. Hence, at least to date, importation of urogenital schistosomiasis to Zanzibar does not seem to be a challenge and threat for elimination efforts. This finding is in contrast to reports from the neighboring island Mafia, where urogenital schistosomiasis is considered as an imported disease [49]. Also, access to clean water by having a tap or well near or at the home was not significantly associated with a *S. haematobium* infection in our study. This is in contrast to assumptions that access to clean water can improve the situation in endemic countries and reduce exposure to schistosomes [42,50]. However, one has to

keep in mind that many drivers for infection are complex, interconnected and multidirectional and that a systems epidemiology approach is necessary to put them into context [51].

A limitation of our large-scale community- and school-based study that involved more than 10,000 people annually is the yet small number of individuals and particularly adults that were diagnosed *S. haematobium*-positive. While this is an expected impediment when working in an elimination setting, the low infection numbers rendered the risk factor assessment difficult and resulted in huge relative differences in the *S. haematobium* prevalence between some years. Moreover, we used basic parasitological methods (reagent strips and urine filtration microscopy) to detect *S. haematobium* infections, which are not very sensitively detecting very light intensity infections [52]. Hence, many more people than identified might have carried a very light intensity infection, thereby biasing associations and resulting in an underestimation of the true prevalence.

Zanzibar has made great achievements towards schistosomiasis elimination over the past years applying population-based interventions. The current evidence that the S. haematobium prevalence and morbidity rebound quickly over the period of a 16-month treatment gap in hotspot areas but remain low in areas with a weaker force of transmission, emphasizes the need to rethink carefully future elimination strategy approaches. Clearly, due to the spatial heterogeneity of S. haematobium infections in Zanzibar and the varying risk of reinfection, future interventions aiming for interruption of S. haematobium transmission need to consider the microepidemiology of the islands and be adapted to it [19]. We hence suggest that interventions targeting hotspot areas in Zanzibar should be re-intensified and ideally include biannual MDA, snail control, behavior change and, in a multi-sectoral approach, improved access to safe water and sanitation. In low prevalence areas, future efforts might shift from MDA towards surveillance-response, including risk-based test-and-treat approaches using new and sensitive pointof-care diagnostic tools, xenomonitoring, focal snail control and health communication to ensure the identification and treatment of those infected but to avoid overtreatment and treatment fatigue of the healthy population. Setting up an effective surveillance-response system in areas that are or become low prevalence areas, will also help to sustain the gains made by reacting to any outbreak and sign of recrudescence in time and thus to ultimately progress towards a sustained interruption of S. haematobium transmission across Zanzibar. Our results also indicate that interruption of MDA and potentially other interventions, for example caused by a lack of donor funding, procurement issues, or the current Covid-19 pandemic, can result in a rapid resurgence of transmission and disease that needs to be reacted to in time. Finally, our study highlights the need for more research on the sustainability of the gains of control and on the criteria and conditions required for stopping schistosomiasis MDA.

Supporting information

S1 Dataset. Data that support the findings of this study from 2018–2020. (CSV)

S1 Dictionary. Data dictionary that explains the data collected from 2018–2020. (TXT)

Acknowledgments

We are grateful to all the schoolchildren and adults who participated in the surveys over the years 2011/12 to 2020. We thank the dedicated field- and laboratory teams in Pemba and Unguja who contributed to this work.

Author Contributions

Conceptualization: Jan Hattendorf, Said Mohammed Ali, Fatma Kabole, David Rollinson, Stefanie Knopp.

Data curation: Lydia Trippler, Stefanie Knopp.

- Formal analysis: Lydia Trippler, Jan Hattendorf, Stefanie Knopp.
- Funding acquisition: David Rollinson, Stefanie Knopp.
- **Investigation:** Lydia Trippler, Shaali Makame Ame, Saleh Juma, Salum Abubakar, Said Mohammed Ali, Fatma Kabole, David Rollinson, Stefanie Knopp.
- **Methodology:** Lydia Trippler, Shaali Makame Ame, Saleh Juma, Salum Abubakar, Said Mohammed Ali, Fatma Kabole, Stefanie Knopp.
- **Project administration:** Shaali Makame Ame, Said Mohammed Ali, Fatma Kabole, David Rollinson, Stefanie Knopp.
- Resources: Said Mohammed Ali, Fatma Kabole, David Rollinson.
- Supervision: Shaali Makame Ame, Said Mohammed Ali, Fatma Kabole, David Rollinson, Stefanie Knopp.
- Visualization: Lydia Trippler, Jan Hattendorf, Stefanie Knopp.
- Writing original draft: Lydia Trippler, Stefanie Knopp.
- Writing review & editing: Lydia Trippler, Shaali Makame Ame, Jan Hattendorf, Saleh Juma, Salum Abubakar, Said Mohammed Ali, Fatma Kabole, David Rollinson, Stefanie Knopp.

References

- GBD 2017 DALYs and HALE Collaborators. Global, regional, and national disability-adjusted life-years (DALYs) for 359 diseases and injuries and healthy life expectancy (HALE) for 195 countries and territories, 1990–2017: a systematic analysis for the Global Burden of Disease Study 2017. Lancet. 2018; 392 (10159):1859–922. Epub 2018/11/13. https://doi.org/10.1016/S0140-6736(18)32335-3 PMID: 30415748
- WHO. Schistosomiasis key facts 2020 [cited 2020 12.04.2020]. Available from: https://www.who.int/ news-room/fact-sheets/detail/schistosomiasis.
- 3. WHO. Ending the neglect to attain the Sustainable Development Goals—A road map for neglected tropical diseases 2021–2030. World Health Organization, 2020.
- Deol AK, Fleming FM, Calvo-Urbano B, Walker M, Bucumi V, Gnandou I, et al. Schistosomiasis assessing progress toward the 2020 and 2025 global goals. N Engl J Med. 2019; 381(26):2519–28. https://doi.org/10.1056/NEJMoa1812165 PMID: 31881138
- Toor J, Rollinson D, Turner HC, Gouvras A, King CH, Medley GF, et al. Achieving elimination as a public health problem for *Schistosoma mansoni* and *S. haematobium*: When is community-wide treatment required? J Infect Dis. 2020; 221(Supplement_5):S525–s30. Epub 2019/12/13. https://doi.org/10.1093/ infdis/jiz609 PMID: 31829414
- Knopp S, Person B, Ame SM, Ali SM, Hattendorf J, Juma S, et al. Evaluation of integrated interventions layered on mass drug administration for urogenital schistosomiasis elimination: a cluster-randomised trial. Lancet Glob Health. 2019; 7(8):e1118–e29. Epub 2019/07/01. https://doi.org/10.1016/S2214-109X(19)30189-5 PMID: 31255591
- Allan F, Ame SM, Tian-Bi YT, Hofkin BV, Webster BL, Diakité NR, et al. Snail-related contributions from the Schistosomiasis Consortium for Operational Research and Evaluation program including xenomonitoring, focal mollusciciding, biological control, and modeling. Am J Trop Med Hyg. 2020; 103 (1_Suppl):66–79. Epub 2020/05/14. https://doi.org/10.4269/ajtmh.19-0831 PMID: 32400353
- Pennance T, Archer J, Lugli EB, Rostron P, Llanwarne F, Ali SM, et al. Development of a molecular snail xenomonitoring assay to detect *Schistosoma haematobium* and *Schistosoma bovis* infections in their *Bulinus* snail hosts. Molecules. 2020; 25(17). Epub 2020/09/06. https://doi.org/10.3390/ molecules25174011 PMID: 32887445

- Gurarie D, Lo NC, Ndeffo-Mbah ML, Durham DP, King CH. The human-snail transmission environment shapes long term schistosomiasis control outcomes: Implications for improving the accuracy of predictive modeling. PLoS Negl Trop Dis. 2018; 12(5):e0006514. Epub 2018/05/22. https://doi.org/10.1371/ journal.pntd.0006514 PMID: 29782500
- Mgeni AF, Kisumku UM, McCullough FS, Dixon H, Yoon SS, Mott KE. Metrifonate in the control of urinary schistosomiasis in Zanzibar. Bull World Health Organ. 1990; 68(6):721–30. Epub 1990/01/01. PMID: 2127381
- Savioli L, Dixon H, Kisumku UM, Mott KE. Control of morbidity due to Schistosoma haematobium on Pemba Island: programme organization and management. Trop Med Parasitol. 1989; 40(2):189–94. Epub 1989/06/01. PMID: 2505381
- Albonico M, Chwaya HM, Montresor A, Stolfzfus RJ, Tielsch JM, Alawi KS, et al. Parasitic infections in Pemba Island school children. East Afr Med J. 1997; 74(5):294–8. Epub 1997/05/01. PMID: 9337006
- Savioli L, Dixon H, Kisumku UM, Mott KE. Control of morbidity due to *Schistosoma haematobium* on Pemba Island; selective population chemotherapy of schoolchildren with haematuria to identify highrisk localities. Trans R Soc Trop Med Hyg. 1989; 83(6):805–10. Epub 1989/11/01. <u>https://doi.org/10. 1016/0035-9203(89)90336-2</u> PMID: 2515637
- Stothard JR, Mook P, Mgeni AF, Khamis IS, Khamis AN, Rollinson D. Control of urinary schistosomiasis on Zanzibar (Unguja Island): a pilot evaluation of the educational impact of the Juma na Kichocho health booklet within primary schools. Mem Inst Oswaldo Cruz. 2006; 101 Suppl 1:119–24. Epub 2007/02/20.
- Stothard JR, French MD, Khamis IS, Basanez MG, Rollinson D. The epidemiology and control of urinary schistosomiasis and soil-transmitted helminthiasis in schoolchildren on Unguja Island, Zanzibar. Trans R Soc Trop Med Hyg. 2009; 103(10):1031–44. Epub 2009/05/05. https://doi.org/10.1016/j.trstmh.2009. 03.024 PMID: 19409588
- Guidi A, Andolina C, Makame Ame S, Albonico M, Cioli D, Juma Haji H. Praziquantel efficacy and longterm appraisal of schistosomiasis control in Pemba Island. Trop Med Int Health. 2010; 15(5):614–8. Epub 2010/03/11. https://doi.org/10.1111/j.1365-3156.2010.02488.x PMID: 20214757
- Knopp S, Mohammed KA, Ali SM, Khamis IS, Ame SM, Albonico M, et al. Study and implementation of urogenital schistosomiasis elimination in Zanzibar (Unguja and Pemba islands) using an integrated multidisciplinary approach. BMC Public Health. 2012; 12:930. Epub 2012/11/01. <u>https://doi.org/10.1186/ 1471-2458-12-930 PMID: 23110494</u>
- Knopp S, Person B, Ame SM, Mohammed KA, Ali SM, Khamis IS, et al. Elimination of schistosomiasis transmission in Zanzibar: baseline findings before the onset of a randomized intervention trial. PLoS Negl Trop Dis. 2013; 7(10):e2474. Epub 2013/10/23. <u>https://doi.org/10.1371/journal.pntd.0002474</u> PMID: 24147165
- Knopp S, Ame SM, Person B, Hattendorf J, Rabone M, Juma S, et al. A 5-Year intervention study on elimination of urogenital schistosomiasis in Zanzibar: Parasitological results of annual cross-sectional surveys. PLoS Negl Trop Dis. 2019; 13(5):e0007268. Epub 2019/05/07. https://doi.org/10.1371/journal. pntd.0007268 PMID: 31059495
- 20. Office of Chief Government Statistician Zanzibar. Zanzibar statistical abstract 2019. 2020.
- 21. Office of Chief Government Statistician Zanzibar. Zanzibar in figures 2019. 2020.
- 22. Office of Chief Government Statistician Zanzibar. Household budget survey 2014/2015. 2016.
- 23. Ministry of Health Tanzania. Tanzania demographic and health survey and malaria indicator survey 2015–2016—Final report. 2016.
- 24. WHO. Prevention and control of schistosomiasis and soil-transmitted helminthiasis: report of a WHO expert committee. World Health Organization, 2002.
- Montresor A, Engels D, Ramsan M, Foum A, Savioli L. Field test of the 'dose pole' for praziquantel in Zanzibar. Trans R Soc Trop Med Hyg. 2002; 96(3):323–4. Epub 2002/08/15. https://doi.org/10.1016/ s0035-9203(02)90111-2 PMID: 12174788
- Montresor A, Crompton D, Hall A, Bundy D, Savioli L. Guidelines for the evaluation of soil-transmitted helminthiasis and schistosomiasis at community level. World Health Organization, 1998.
- 27. Krauth SJ, Greter H, Stete K, Coulibaly JT, Traoré SI, Ngandolo BN, et al. All that is blood is not schistosomiasis: experiences with reagent strip testing for urogenital schistosomiasis with special consideration to very-low prevalence settings. Parasit Vectors. 2015; 8:584. Epub 2015/11/12. https://doi.org/ 10.1186/s13071-015-1165-y PMID: 26554822
- Tomson C, Porter T. Asymptomatic microscopic or dipstick haematuria in adults: which investigations for which patients? A review of the evidence. BJU Int. 2002; 90(3):185–98. Epub 2002/07/23. https://doi.org/10.1046/j.1464-410x.2002.02841.x PMID: 12133052
- McDonald MM, Swagerty D, Wetzel L. Assessment of microscopic hematuria in adults. Am Fam Physician. 2006; 73(10):1748–54. Epub 2006/06/01. PMID: 16734050

- Kittur N, King CH, Campbell CH, Kinung'hi S, Mwinzi PNM, Karanja DMS, et al. Persistent hotspots in Schistosomiasis Consortium for Operational Research and Evaluation studies for gaining and sustaining control of schistosomiasis after four years of mass drug administration of praziquantel. Am J Trop Med Hyg. 2019; 101(3):617–27. Epub 2019/07/10. <u>https://doi.org/10.4269/ajtmh.19-0193</u> PMID: 31287046
- Wiegand RE, Mwinzi PNM, Montgomery SP, Chan YL, Andiego K, Omedo M, et al. A persistent hotspot of *Schistosoma mansoni* infection in a five-year randomized trial of praziguantel preventative chemotherapy strategies. J Infect Dis. 2017; 216(11):1425–33. Epub 2017/10/03. <u>https://doi.org/10.1093/</u> infdis/jix496 PMID: 28968877
- Elmorshedy H, Bergquist R, El-Ela NE, Eassa SM, Elsakka EE, Barakat R. Can human schistosomiasis mansoni control be sustained in high-risk transmission foci in Egypt? Parasit Vectors. 2015; 8:372. Epub 2015/07/16. https://doi.org/10.1186/s13071-015-0983-2 PMID: 26174621
- 33. Dejon-Agobé JC, Edoa JR, Honkpehedji YJ, Zinsou JF, Adégbitè BR, Ngwese MM, et al. Schistosoma haematobium infection morbidity, praziquantel effectiveness and reinfection rate among children and young adults in Gabon. Parasit Vectors. 2019; 12(1):577. <u>https://doi.org/10.1186/s13071-019-3836-6</u> PMID: 31823806
- 34. Chadeka EA, Nagi S, Sunahara T, Cheruiyot NB, Bahati F, Ozeki Y, et al. Spatial distribution and risk factors of *Schistosoma haematobium* and hookworm infections among schoolchildren in Kwale, Kenya. PLoS Negl Trop Dis. 2017; 11(9):e0005872–e. <u>https://doi.org/10.1371/journal.pntd.0005872</u> PMID: 28863133
- 35. Kittur N, Binder S, Campbell CH, King CH, Kinung'hi S, Olsen A, et al. Defining persistent hotspots: areas that fail to decrease meaningfully in prevalence after multiple years of mass drug administration with praziquantel for control of schistosomiasis. Am J Trop Med Hyg. 2017; 97(6):1810–7. Epub 2017/ 10/11. https://doi.org/10.4269/ajtmh.17-0368 PMID: 29016344
- Mari L, Ciddio M, Casagrandi R, Perez-Saez J, Bertuzzo E, Rinaldo A, et al. Heterogeneity in schistosomiasis transmission dynamics. J Theor Biol. 2017; 432:87–99. <u>https://doi.org/10.1016/j.jtbi.2017.08</u>. 015 PMID: 28823529
- Pennance T, Person B, Muhsin MA, Khamis AN, Muhsin J, Khamis IS, et al. Urogenital schistosomiasis transmission on Unguja Island, Zanzibar: Characterisation of persistent hot-spots. Parasit Vectors. 2016; 9(1):646. Epub 2016/12/18. https://doi.org/10.1186/s13071-016-1847-0 PMID: 27986092
- Senghor B, Diaw OT, Doucoure S, Sylla SN, Seye M, Talla I, et al. Efficacy of praziquantel against urinary schistosomiasis and reinfection in Senegalese school children where there is a single well-defined transmission period. Parasit Vectors. 2015; 8(1):362. <u>https://doi.org/10.1186/s13071-015-0980-5</u> PMID: 26156522
- Gryseels B, Polman K, Clerinx J, Kestens L. Human schistosomiasis. Lancet. 2006; 368(9541):1106– 18. Epub 2006/09/26. https://doi.org/10.1016/S0140-6736(06)69440-3 PMID: 16997665
- 40. Colley DG, Secor WE. Immunology of human schistosomiasis. Parasite Immunol. 2014; 36(8):347–57. Epub 2014/08/22. https://doi.org/10.1111/pim.12087 PMID: 25142505
- Lee Y-H, Lee J-S, Jeoung H-G, Kwon I-S, Mohamed AAWS, Hong S-T. Epidemiological survey on schistosomiasis and intestinal helminthiasis among village residents of the rural river basin area in White Nile State, Sudan. Korean J Parasitol. 2019; 57(2):135–44. Epub 2019/04/30. https://doi.org/10. 3347/kjp.2019.57.2.135 PMID: 31104405
- Kapito-Tembo AP, Mwapasa V, Meshnick SR, Samanyika Y, Banda D, Bowie C, et al. Prevalence distribution and risk factors for *Schistosoma hematobium* infection among school children in Blantyre, Malawi. PLoS Negl Trop Dis. 2009; 3(1):e361. <u>https://doi.org/10.1371/journal.pntd.0000361</u> PMID: 19156193
- 43. Nagi S, Chadeka EA, Sunahara T, Mutungi F, Justin YK, Kaneko S, et al. Risk factors and spatial distribution of *Schistosoma mansoni* infection among primary school children in Mbita District, Western Kenya. PLoS Negl Trop Dis. 2014; 8(7):e2991. Epub 2014/07/25. https://doi.org/10.1371/journal.pntd. 0002991 PMID: 25058653
- Atalabi TE, Lawal U, Ipinlaye SJ. Prevalence and intensity of genito-urinary schistosomiasis and associated risk factors among junior high school students in two local government areas around Zobe Dam in Katsina State, Nigeria. Parasit Vectors. 2016; 9(1):388. https://doi.org/10.1186/s13071-016-1672-5 PMID: 27388007
- 45. Rudge JW, Stothard JR, Basanez MG, Mgeni AF, Khamis IS, Khamis AN, et al. Micro-epidemiology of urinary schistosomiasis in Zanzibar: Local risk factors associated with distribution of infections among schoolchildren and relevance for control. Acta Trop. 2008; 105(1):45–54. Epub 2007/11/13. https://doi. org/10.1016/j.actatropica.2007.09.006 PMID: 17996207
- 46. Geleta S, Alemu A, Getie S, Mekonnen Z, Erko B. Prevalence of urinary schistosomiasis and associated risk factors among Abobo Primary School children in Gambella Regional State, southwestern

Ethiopia: a cross sectional study. Parasit Vectors. 2015; 8(1):215. https://doi.org/10.1186/s13071-015-0822-5 PMID: 25886292

- Angora EK, Boissier J, Menan H, Rey O, Tuo K, Touré AO, et al. Prevalence and risk factors for schistosomiasis among schoolchildren in two settings of Côte d'Ivoire. Trop Med Infect Dis. 2019; 4(3):110. https://doi.org/10.3390/tropicalmed4030110 PMID: 31340504
- 48. Person B, Ali SM, A'Kadir FM, Ali JN, Mohammed UA, Mohammed KA, et al. Community knowledge, perceptions, and practices associated with urogenital schistosomiasis among school-aged children in Zanzibar, United Republic of Tanzania. PLoS Negl Trop Dis. 2016; 10(7):e0004814. Epub 2016/07/12. https://doi.org/10.1371/journal.pntd.0004814 PMID: 27399310
- 49. Stothard JR, Ameri H, Khamis IS, Blair L, Nyandindi US, Kane RA, et al. Parasitological and malacological surveys reveal urogenital schistosomiasis on Mafia Island, Tanzania to be an imported infection. Acta Trop. 2013; 128(2):326–33. Epub 2012/09/27. <u>https://doi.org/10.1016/j.actatropica.2012.09.006</u> PMID: 23010161
- Grimes JET, Croll D, Harrison WE, Utzinger J, Freeman MC, Templeton MR. The relationship between water, sanitation and schistosomiasis: A systematic review and meta-analysis. PLoS Negl Trop Dis. 2014; 8(12):e3296. https://doi.org/10.1371/journal.pntd.0003296 PMID: 25474705
- 51. Krauth SJ, Balen J, Gobert GN, Lamberton PHL. A call for systems epidemiology to tackle the complexity of schistosomiasis, its control, and its elimination. Trop Med Infect Dis. 2019; 4(1):21.
- 52. Knopp S, Ame SM, Hattendorf J, Ali SM, Khamis IS, Bakar F, et al. Urogenital schistosomiasis elimination in Zanzibar: accuracy of urine filtration and haematuria reagent strips for diagnosing light intensity *Schistosoma haematobium* infections. Parasit Vectors. 2018; 11(1):552. Epub 2018/10/26. <u>https://doi.org/10.1186/s13071-018-3136-6</u> PMID: 30352631